Author's response to reviews

Title: Anti-Angiotensin Converting Enzyme (ACE) Proteins from Mycelia of Ganoderma lucidum (Curtis) P. Karst

Authors:

Nurhuda Mohamad Ansor (huda_ansor@yahoo.com)
Noorlidah Abdullah (noorlidah@um.edu.my)
Norhaniza Aminudin (hanizaaminudin@um.edu.my)

Version: 2 Date: 11 September 2013

Author's response to reviews: see over
10th September 2013

To
The Editor
BMC Complementary and Alternative Medicine

Dear Sir/Madam,

Thank you for your useful comments on our manuscript ‘Anti-angiotensin converting enzyme (ACE) proteins from mycelia of Ganoderma lucidum (Curtis) P. Karst’. We have modified the manuscript accordingly, and detailed corrections are listed below point by point:

1) Figure of SDS PAGE needs to be given in 'Results and discussion'.
   We have inserted the figure of SDS PAGE (Figure 3) in ‘Results and discussion’.

2) Page No. 11, 3rd para, second line: Sentence needs to be corrected for grammatical error.
   Correction has been made on the particular sentence.

3) In 'Conclusions' - More clarity with correction of grammatical errors are required.
   We have made appropriate modifications in ‘Conclusions’.

4) A careful revision is required before re-submission of the paper.
   The paper has been carefully revised for re-submission.

5) Needs some language corrections before being published.
   We have sent the manuscript to professional language editing service for proofreading.

The manuscript has been resubmitted to your journal. We look forward to your positive response.

Sincerely and with regards,

Norhaniza Aminudin, PhD.
Senior lecturer
Institute of Biological Science, Faculty of Science
University of Malaya, 50603 Kuala Lumpur
Malaysia.